CENWP-OP-B                                                                                                                                  11 July 2005

MEMORANDUM FOR CENWP-OP, ATTN: Randy Bailey

SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for Week 28 of 2005, which covers the period from 03 through 09 July 2005.

1.  OPERATION SUMMARY

1.1 Daily average river flows ranged from 153.8 to 196.3 kcfs.  Daily average project tailwater elevation ranged from 13.9’ to 16.5’ msl.  Daily average powerhouse forebay elevation ranged from 73.8’ to 75.2’ msl.  Daily average river temperature ranged from 66 to 67(F.  Secchi disk measurements ranged from 4.8 to 6.5 feet.

1.2 Spill Bays 1 and 18, which supply attraction flow for Cascades Island and B-Branch respectively, are included in the spill pattern.  Daily average spill ranged from 84.3 to 97.1 kcfs.   

1.3 Unit operation.   

1.3.1 Unit 0 remains out of service due to governor trouble.

1.3.2 Unit 2 remains out of service for turbine rehab. 

1.3.3 Unit 10 remains out of service for gantry crane three construction.

1.3.4 Unit 15 was out of service from 0030 until 0527 on 09 July due to loss of line protection.  

1.3.5 Unit 16 went out of service at 2255 on 08 July for governor control work.

1.3.6 Unit 17 was out of service from 0446 until 1110 on 04 July due to water in the turbine bearing.  Was out of service from 0030 until 0526 on 09 July due to loss of line protection.   

1.3.7 Unit 18 was out of service from 2330 on 08 July until 0523 on 09 July due to loss of line protection.

1.3.8 Fish Unit 2 was out of service from 2338 on 08 July until 0120 on 09 July due to loss of line protection.  Out of service from 0452 until 1000 on 09 July for ZJ switching.  

1.4 Second Powerhouse fish units provide attraction flow for the Washington Shore fish ladder.  The units are placed on standby from approximately 0000-0300 nightly if drawdown exceeds 1.0’.
1.5 Gatewell drawdowns are measured on Wednesdays.  Drawdowns are measured daily on Unit 17.

2.  ACTIVITIES

2.1 AWS Closures.  The following AWS valves were closed this reporting week.  

2.1.1 FV1-1 closed for cleaning from 1506 until 1518 on 07 July.

2.1.2 FV3-9 closed for cleaning from 1459 until 1505 on 07 July.

2.2 STS/VBS Inspections.   No STSs/VBSs were inspected this reporting week.  

2.3 Dewatering and Fish Salvage.  No dewatering or salvage activities occurred this reporting week.  
3.  RESEARCH
3.1 Battelle FPE and FGE.  Monitoring continues.

3.2 CRITFC.  Sampling continues twice a week through the summer.

3.3 CTUIR.  Received 100 lamprey from NOAA Fisheries on 16 and 22 June.

3.4 ERDC TDG.  Sensors remain installed in the tailrace.

3.5 NOAA Fish Lamprey.   The NDE ramp installation is currently on hold. 

3.6 PSMFC.   Zero fallbacks were observed crossing the separator bars this week.  Subyearling chinook research fish collection: 1070 for USGS and 265 for NMFS.  
3.7 Univ of Idaho.  Salmon trapping concluded until 01 August.  Lamprey trapping and testing continues.
3.8 USDA Wildlife Services.   Hazing activities for gulls continue.

3.9 USGS FPE and Survival.  Juvenile tracking continues.  
3.10 Warm Springs Tribes.  Received 49 lamprey from NOAA Fisheries on 29 June.
3.11  WDFW Fish Counting.  16-hr visual counting at Washington Shore and Bradford Island. 

4.  FISHWAYS

4.1 Project biologists inspected 03 July through 09 July.  
4.2 Fish Passage Plan Observations.

	DATE
	LOCATion
	FPP Deviation
	CAUSE
	RESPONSE
	TR#

	All week
	PH1 N PCC
	Velocity <1.5 fps
	Slow water
	Normal condition.
	

	All week
	PH2 SCCH
	Velocity <1.5 fps
	Improper install.
	Repairs scheduled for 06/07
	

	All week
	Cascades Is.
	FG6-12 OOS
	Bent shaft
	See 6.1.2
	

	7/3-8
	DSM2 12A-S
	Closed/inoperable
	Rod bolt broken
	Operators submitted TR
	

	7/6-9
	B-Branch
	North entrance gate closed.
	Separated from shaft
	Mechanics to try to repair on 11 July.
	


Table 2.  FPP Out of Criteria items.

4.3 Fish Ladder Temperatures.  Measurements are taken three weirs down from the count station at Bradford Island as well as at the junction of the UMT and the main ladder at Washington shore.   The AFF is measured inside the lab at the fish return.  

	Date
	WA Shore

Average
	Bradford Is

Average
	AFF exit max

	7/03
	66.2
	66.5
	66.3

	7/04
	66.4
	66.5
	66.4

	7/05
	66.9
	66.8
	66.9

	7/06
	67.1
	67.2
	67.3

	7/07
	67.4
	67.4
	67.5

	7/08
	67.4
	67.4
	67.4

	7/09
	67.1
	67.1
	67.2


Table 3.  Bonneville Fishway temperatures.

4.4 Sensor Calibration.  Calibration was checked on 09 July.

4.5 Zebra Mussel Monitoring.  No signs of colonization were observed this reporting week.  

5.   CONSTRUCTION

5.1 B2CC PIT Tag Detector.  Installation is planned for winter 05/06.

5.2 B2 Fish Unit Trash rake.  Delivery and installation scheduled for late summer 2005.

5.3 B2 VBS modifications.  Gatewell modifications are scheduled to begin in winter 05/06.

5.4 Bradford Island Adult PIT Tag Detectors.  Installation scheduled for winter 05/06.

5.5 PH 1 Gantry Crane Delivery and Erection.  Nothing new to report.

5.6 PH1 STS/VBS removal and TIES removal.  Six TIES and 23 PH1 STSs/VBSs will be removed in August.

5.7 SMF Full flow PIT Tag Detectors.  CH2M Hill has been contracted to write the letter report.

6.  FISHWAY STATUS

6.1  Adult Fishways.    
6.1.1 B-Branch and Cascades Island fishway PLCs are in manual.  B-branch entrance and ladder transducers were replaced and calibrated this week.
6.1.2 Cascades Island diffuser FG6-12 has a bent shaft and will remain out of service until winter 2006/07.  FG6-11 is used to compensate.
6.1.3 A-branch has been reprogrammed.  FG3-4 will remain out of service until winter 2005/06.
6.1.4 PH1 NCCH sensor still OOS.  Channel elevation is now controlled from CCCH.
6.1.5 SLEDS remain in the out position until further notice due to structural damage.  
6.2 Juvenile Fishways. 
6.2.1 DSM2 was out of criteria from 0855 until 0919 on 07 July.  Please see 8.3
7.  HAZMAT, SPILLS AND CLEANUP – Nothing to report this week.
8.  TELETYPES AND OTHER NOTEWORTHY ITEMS.
8.1 A teletype was issued on 01 July requesting that Bonneville pool be operated between elevation 74.5’ – 76.0’ for treaty gillnet fishing effective 0600 hrs Tuesday, 05 Jul 2005 - 1800 hrs Thurs, 07 Jul 2005.

8.2 Teletypes issued on 05 July and 06 July requested the project to exceed the normal summertime 24-hour tailwater fluctuation limit of 4.0’.  Tailwater was permitted to fluctuate up to 5.0’ from 1700 on 05 July until 1700 on 06 July and from 2100 on 06 July until 2100 on 07 July.  The maximum rate of change is 1.5’ per hour.  This is to maintain required forebay elevations for the tribal gillnet fishery.

8.3 At 0855, electricians attempted to work on the DSM2 PLC.  They turned the PLC to off before first putting the system in manual.  As a result, the ERG opened and the DSM went out of fish passage criteria.  The SMF crew rescued about five smolts from the PDS before the system could be brought back into criteria.  The DSM was back to normal by 0919.  In the future, fisheries will first be notified of any work on fish passage related equipment, and appropriate precautions will be taken to avoid this type of situation from happening again.
8.4 A teletype issued on 07 July allows Bonneville to calibrate a new sensor for spill bay 15.  The gate will be opened and closed during calibrations.  Changes in spill volumes will be compensated with bays 14 and 16.
JAMES R. MAHAR, P.E.

Operations Manager
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